Author's response to reviews

Title: Genotype-independent association between profound vitamin D deficiency and delayed sputum smear conversion in pulmonary tuberculosis

Authors:

Kashaf Junaid (kashaf_junaid@hotmail.com)
Abdul Rehman (rehman_mmg@hotmail.com)
Tahir Saeed (doctor_tahir@hotmail.com)
David A Jolliffe (d.a.jolliffe@qmul.ac.uk)
Kristie Wood (kristie.wood@qmul.ac.uk)
Adrian R Martineau (a.martineau@qmul.ac.uk)

Version: 2
Date: 22 October 2014

Author's response to reviews:

we have revised the title of our manuscript that is more descriptive to our study